%&LaTeX
\documentclass{article}

\usepackage[utf8]{inputenc}
\usepackage[T1]{fontenc}
\usepackage{textcomp}

\begin{document}

\section*{2006}

Anton Cervantes, Gemma Sanchez, Josep Llados, Agnes Borras, \& Ana Rodriguez. (2006).<a target={\textquoteright}_blank{\textquoteright} href={\textquoteright}/files/CSL2006.pdf{\textquoteright}> "Biometric Recognition Based on Line Shape Descriptors "</a> In \textit{Lecture Notes in Computer Science} (Vol. 3926, pp. 346--357,). Springer Link.

\end{document}

